

Challenges in Preparing and Implementing a Clinical Trial at Field Level in an Ebola Emergency: A Case Study in Guinea, West Africa

Authors	Carazo Perez, S; Folkesson, E; Anglaret, X; Beavogui, A; Berbain, E; Camara, A; Depoortere, E; Lefevre, A; Maes, P; Malme, K; Malvy, J; Ombelet, S; Poelaert, G; Sissoko, D; Tounkara, A; Trbovic, P; Piguet, P; Antierens, A
Citation	Challenges in Preparing and Implementing a Clinical Trial at Field Level in an Ebola Emergency: A Case Study in Guinea, West Africa. 2017, 11 (6):e0005545 PLoS Negl Trop Dis
DOI	10.1371/journal.pntd.0005545
Publisher	Public Library of Science
Journal	PLoS Neglected Tropical Diseases
Rights	Archived with thanks to PLoS Neglected Tropical Diseases
Download date	03/10/2021 16:52:50
Link to Item	http://hdl.handle.net/10144/618954



Citation: Carazo Perez S, Folkesson E, Anglaret X, Beavogui A-H, Berbain E, Camara A-M, et al. (2017) Challenges in preparing and implementing a clinical trial at field level in an Ebola emergency: A case study in Guinea, West Africa. PLoS Negl Trop Dis 11(6): e0005545. https://doi.org/10.1371/ journal.pntd.0005545

Editor: Manuel Schibler, University of Geneva Hospitals, SWITZERLAND

Published: June 22, 2017

Copyright: © 2017 Carazo Perez et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper.

Funding: The authors received no specific funding for this work.

Competing interests: The authors have declared that no competing interests exist.

VIEWPOINTS

Challenges in preparing and implementing a clinical trial at field level in an Ebola emergency: A case study in Guinea, West Africa

Sara Carazo Perez^{1,2}*, Elin Folkesson¹, Xavier Anglaret^{3,4}, Abdoul-Habib Beavogui⁵, Emmanuel Berbain¹, Alseny-Modet Camara⁶, Evelyn Depoortere^{1,7}, Annabelle Lefevre¹, Piet Maes^{8,9}, Kristian Nødtvedt Malme¹, Jean-Marie Denis Malvy^{3,10}, Sien Ombelet¹, Geertrui Poelaert¹, Daouda Sissoko^{3,10}, Alexis Tounkara⁶, Pierre Trbovic¹, Pascal Piguet¹, Annick Antierens¹*

 Médecins Sans Frontières, Brussels, Belgium, 2 Department of Preventive and Social Medicine, Laval University, Québec City, Canada, 3 Inserm, UMR 1219, Université de Bordeaux, Bordeaux, France,
Programme PACCI, Abidjan, Côte d'Ivoire, 5 Centre de Formation et de Recherche en Santé Rurale de Mafèrinyah, Conackry, Guinea, 6 Médecins Sans Frontières, Conackry, Guinea, 7 Institute of Tropical Medicine, Antwerpen, Belgium, 8 European Mobile Laboratory Project, Hamburg, Germany, 9 KU Leuven, Rega Institute for Medical Research, Leuven, Belgium, 10 Pellegrin Hospital, Bordeaux, France

* sara.carazo@gmail.com (SCP); annick.antierens@geneva.msf.org (AA)

Author summary

During the large Ebola outbreak that affected West Africa in 2014 and 2015, studies were launched to evaluate potential treatments for the disease. A clinical trial to evaluate the effectiveness of the antiviral drug favipiravir was conducted in Guinea. This paper describes the main challenges of the implementation of the trial in the Ebola treatment center of Guéckédou. Following the principles of the Good Clinical Research Practices, we explored the aspects of the community's communication and engagement, ethical conduct, trial protocol compliance, informed consent of participants, ongoing benefit/risk assessment, record keeping, confidentiality of patients and study data, and roles and responsibilities of the actors involved. We concluded that several challenges have to be addressed to successfully implement a clinical trial during an international medical emergency but that the potential for collaboration between research teams and humanitarian organizations needs to be highlighted.

Introduction

The largest and most complex Ebola virus outbreak in history started in Guinea in December 2013 and spread rapidly to neighboring countries in March 2014 [1]. As no effective treatment existed against Ebola virus disease (EVD), Ebola treatment centers (ETC) were only able to provide supportive care with case-fatality rates above 50% [2].

In August 2014, the World Health Organization (WHO) convened an advisory panel that unanimously concluded it would be acceptable to assess potential unregistered interventions that had shown promising results in preclinical laboratory and animal phases but had not been fully evaluated in phase I or II trials in humans [3]. Médecins Sans Frontières (MSF), which was extensively involved in frontline care for Ebola patients, the French National Research Institute for Health and Medical Research (INSERM), and Guinean authorities created a partnership to conduct a "proof of concept" study on the efficacy of high-dose favipiravir in reducing mortality in patients with EVD. Favipiravir, a novel viral RNA polymerase inhibitor developed by the Japanese company Toyama/Fujifilm, was a prioritized drug on the WHO expert panel's list because of its good human safety data and promising results against the Ebola virus in preclinical studies [4–8]. The multisite JIKI study, named after "hope" in the Malinke language, was a noncomparative, single-arm, open-label clinical trial [9]. Any patient aged 1 year or older with laboratory-confirmed EVD was offered the opportunity to participate in the trial and was asked for informed consent. The European Mobile Laboratory consortium (EMLAB) performed routine laboratory tests (Ebola virus PCR and biochemistry). The protocol of the trial has been described elsewhere [9].

With the recruitment of the first patient in the Guéckédou ETC on December 17, 2014, the JIKI trial became the first phase II study on EVD ever implemented by a research team in collaboration with an international nongovernmental organization. This unique situation was complicated by the international humanitarian crisis associated with the uncontrolled Ebola virus epidemic and a vulnerable resource-poor setting of rural Guinea.

This paper is a systematic assessment of the JIKI trial implementation in the field at the Guéckédou ETC, the first center involved in the study. It does not address the prefield aspects (methodological choices, writing of the protocol). As all authors of the present paper were involved in the trial, the assessment may be subjective. Our goal is to offer the scientific and clinical research communities critical feedback into the issues and challenges of this unique situation in order to highlight potential problems in future EVD outbreaks.

Issues addressed

We will discuss the issues surrounding the implementation of the JIKI trial in the field in order to address the interaction between Good Clinical Research Practices [10] and the significant practical challenges in this particular context (Table 1).

Communication with the community

Communities from Forested Guinea were known to mistrust government and Western humanitarian aid agencies, which led to serious security incidents [11]. EVD was still unknown and many rumors were circulating such as that the disease did not exist; that it had been introduced purposely to reduce the population; and that ETC patients were experimented on, killed, or had their organs harvested [12, 13]. Rumors had to be countered, and the messages to the population on the trial objectives and implementation needed to be comprehensive and adapted to the context to minimize fear and rejection as well as inappropriate expectations. As the spread of the epidemic was unpredictable and a large region potentially affected, numerous villages had to be reached in a very short time.

To ensure the community involvement in the trial, local leaders, including women's organizations, youth groups, religious leaders, and village elders, were invited by the research partners to participate in a Community Advisory Board (CAB). This was headed by the regional coordinator of the fight against Ebola and specifically created to counsel and support the trial implementation. During the preparation phase, the CAB discussed the trial protocol and the communication strategy for the community. Following their recommendations, the focus for communication was put on providing correct and comprehensive information to patients, their families, and all 350 MSF staff members, most of whom originated from Guéckédou prefecture, rather than informing entire communities. Anthropologists and key national staff

Principle	Recommendation
Communication with community	Pretrial recruitment and engagement of community leaders in the trial planning and launch. Thoughtful preparation and pretest of messages about the trial.
Ethical conduct	Evaluation of the inclusion criteria based on equity and risk/benefit assessment.
Protocol compliance	Consensus among clinicians to adapt previous protocols to standard operational procedures (SOPs). Ongoing evaluation of acceptability of changes in patient management to accomplish with trial protocol.
Informed consent	Pretrial assessment of cultural adaptation of informed consent. Continuous training of specific staff to obtain informed consent. Local consultation for handling exceptional cases (e.g., unaccompanied children).
Ongoing benefit/risk assessment	Interpretation of laboratory results considering Ebola virus disease (EVD) infectivity. Daily analysis of mortality data.
Records	Innovative management of data recording in the Ebola virus high-risk setting: installation of a scanner, pictures of files, devices allowing decontamination.
Confidentiality	Communication of results as soon as possible to avoid rumors. Clear and early identification of communication channels with media and political leaders.
Roles and responsibilities	Clear definition of responsibilities and communication channels between all actors involved in the implementation (the French National Research Institute for Health and Medical Research [INSERM], Médecins Sans Frontières [MSF], Guinean authorities, the European Mobile Laboratory consortium [EMLAB]).

Table 1. Key recommendations for a trial implementation in an Ebola outbreak.

https://doi.org/10.1371/journal.pntd.0005545.t001

collaborated to develop the form and content of the messages, which was tested on various members of the staff to ensure correct translation into local languages, sensitivity to local culture, and ease of understanding.

At the same time, a mobile health promotion team addressed rumors in the communities, countered misconceptions, and answered questions on the trial. They also relayed community responses to the trial team to modify the messages to patients and their families. For example, when a rumor arose that 4 EVD survivors treated with favipiravir became "crazy" after returning to their villages, a combined MSF and INSERM team visited the villages and discussed the issue with the families and the local health authorities.

However, the interventions were limited by the rapid geographical expansion of the epidemic. It was not possible to ensure that all villages in the ETC catchment area were included in the communications. Patients referred from prefectures outside Guéckédou would not have heard any information about the trial.

Ethical conduct

Research with human subjects must ensure that 3 basic ethical principles are followed: respect for persons, beneficence, and justice [10, 14, 15]. Favipiravir was already shown to be a promising drug with the potential to improve EVD prognosis; it was therefore considered acceptable to evaluate this drug in EVD patients. Given the high mortality in children under 5 years [16, 17] and the fact that Ebola virus often strikes in clusters, including relatives [18], excluding children would certainly have compromised trial acceptability. Dose-finding or tolerance data in children were not available. A team of paediatric experts concluded that full maturation of the enzymes involved in the metabolism of favipiravir was achieved by 12 months. Weightbased doses were chosen based on adult dosages, for which safety data were available [19]. Children under the age of 1 year or weighing less than 10 kg had to be excluded; however, in practice, this resulted in the exclusion of only 2 infants.

Pregnant women were another particularly vulnerable group. Since no human studies of favipiravir in pregnancy had ever been performed and embryonic teratogenicity was reported in mice, no insurance company would cover the risk for pregnancy; thus, pregnant women had to be excluded from the JIKI trial. Pregnant women, however, had a higher case-fatality rate compared to nonpregnant women. The vast majority of pregnancies in EVD patients resulted in miscarriage or intrauterine death, and the few babies born died in a few days [20, 21].

To avoid denying pregnant women access to a potentially beneficial investigational product, MSF accepted responsibility and liability for the emergency use of favipiravir in this group. A protocol for monitored emergency use, including informed consent and rigorous data collection, was submitted for approval to the appropriate ethics committees. Pregnancy testing was introduced for all women of childbearing age admitted in the ETC.

Nonetheless, a time complication occurred, as the emergency-use protocol for pregnant women was still under ethics review when the trial started. This did not, however, have any practical impact, because no pregnant women were admitted in the ETC during the trial period.

Protocol compliance

The trial's standard operational procedures (SOPs) had to be incorporated into the MSF Ebola virus case management protocol without creating any negative impact on patient care or adding extra risks for the staff. To ensure compatibility with the routine management protocols already in place, the research team wrote the SOPs on site with MSF medical staff. Feasibility and organizational changes were evaluated before implementation to ensure smooth integration of the trial into clinical activities.

After trial initiation, the first challenge was related to the study endpoint, which was survival to day 14. Some patients met MSF's discharge criteria (negative PCR results after 3 days without symptoms) before day 14 but still needed monitoring. Thus, a rehabilitation zone was built so that these patients could move more freely but still remain under surveillance of clinical staff.

The second challenge was the follow-up tests for the trial participants. Routine clinical procedures called for PCR testing at admission and at the end of symptoms, with no blood collection between these 2 samples. However, the trial protocol demanded extra sampling at day 2 and day 4, which increased risks for patients and staff. In practice, the additional venous sampling was considered acceptable as it included follow-up biochemical tests, which were beneficial to patient care. Patients were informed of the increased number of blood draws at the time of informed consent. The volume of each blood collection was limited to a minimum. The diagnostic laboratory adapted its organization to the new requirements, including new schedules and the incorporation of biochemical testing. The clinical staff was trained to use the biochemistry results in EVD practice. There were, however, some limitations related to these new procedures. Adding new tests at the laboratory led to some unexpected difficulties. The platform, Abaxis Piccolo Xpress analyzer, was sensitive to heat, and the laboratory did not have a cooling system. Disinfection of the machine was also a challenge, as the chlorine solution normally used harmed the electronic components.

The third challenge was the need for early initiation of treatment. As the antiviral medication was expected to be more effective the earlier in the course of the disease it was started, it was crucial to include patients in the trial as soon as possible. Treatment initiation, however, could only be done after the laboratory confirmed EVD and informed consent was obtained. This required important changes in the staff shifts to ensure that inclusion could be done as late as 1:00 AM. Moreover, the samples for PCR testing had to reach the diagnostic lab before 6:00 PM, due to the long process of decontaminating samples and performing the necessary tests. Consequently, some patients arriving late in the evening, mostly those transferred from other prefectures, had to wait almost 24 hours before inclusion.

Informed consent

Closely related to the principle of respect for the dignity of participants, an informed consent had to be obtained from all patients before their inclusion in the trial [10]. During the JIKI trial, communication with potential participants was complicated by a setting of fear, mistrust, and the disease itself, as well as by clinical trial illiteracy. As consent was only requested from laboratory-confirmed Ebola virus patients, it had to be obtained in the ETC high-risk zone from medical staff wearing full personal protective equipment, which clearly hampered communication and the perception of free choice for the patient.

Some patients could not decide for themselves without extensive family consultation. In addition, there were no clear legal or traditional rules for ethically obtaining proxy consent for the inclusion of children or impaired persons who arrived at the ETC without guardians.

Several actions were taken to try to ensure a freely given and culturally adapted informed consent. Firstly, a multidisciplinary team of medical, health-promotion, psychosocial, anthropological, and research personnel translated the informed consent into comprehensive and culturally sensitive local languages. Secondly, oral consent was preferred over signed consent because of cultural acceptability; according to the local psychosocial team, signing a document while hospitalized was considered as signing a legal will. Therefore, informed oral consent was signed off by 2 witnesses or by the guardian in the case of minors.

Finally, the Guéckédou CAB was consulted regarding unaccompanied children and advised to seek special authorization from national authorities. The national coordinator of the fight against Ebola was contacted and accepted the responsibility for the inclusion of unaccompanied minor children in the trial.

However, despite efforts to explain the protocol in local languages, it was difficult to assess the amount of information understood by each patient, due to language and cultural barriers and the clinical status of patients.

Ongoing benefit/risk assessment

In order to assure that the benefits continuously outweighed the potential risks associated with the investigational drug, strict pharmacovigilance of favipiravir was crucial, as no safety data were available for the high doses used in the trial. However, due to the severity of EVD and limited availability of clinical facilities, distinguishing potential adverse effects of the favipiravir treatment from the signs and symptoms of EVD was challenging.

During and after administration of favipiravir, the patients were closely monitored for any adverse signs or symptoms not corresponding to typical EVD symptoms, with a focus on those adverse events previously described in clinical studies with favipiravir for the treatment of influenza infection. Biochemical tests were used to observe possible liver and kidney damage. However, at the individual patient level, the interpretation of the causes of deterioration was still difficult, because some signs and symptoms, such as diarrhea, vomiting, or dizziness, could be related to either EVD or favipiravir.

Additionally, mortality data were sent to the Data Safety Monitoring Board (DSMB) on a daily basis. The protocol included rules that allowed for stopping the trial if interim analysis, planned after every 20 patients, showed that on-trial mortality did not trend lower than pretrial mortality.

Records

The quality of the data collection is related to clinical staff qualifications as well as quality of the recording system. Therefore, medical personnel were trained on specific data-collection methods. What proved to be more challenging was handling these data. Due to the infection-control restrictions in the ETC, paper-recorded patient data could not leave the high-risk zone. To address this problem, case report forms and routine patient documents were scanned inside the high-risk zone, and the data were sent electronically to both the medical and research databases. Compared to the pretrial practice of verbal transmission of patient data between the high- and low-risk zones, this new procedure resulted in fewer transcription mistakes and less time for the staff in the high-risk zone.

Confidentiality

The context of the trial, with high social, media, and political pressure, threatened the confidentiality of the study data. The trial was an open-label, single-arm study. After the trial was launched, most EVD-confirmed patients admitted into the ETC received favipiravir treatment. This was known by the staff and easy to grasp for the local communities. As the primary study outcome was mortality, confidentiality in regards to individual patient outcome and the potential efficacy of the favipiravir treatment was difficult to keep undisclosed to the outside world. In an attempt to limit the spread of rumors in the local community, regular meetings with the staff were held to explain the need for prudent interpretation of the observed outcomes and the importance of confidentiality regarding patients and trial results. The ETC teams and CAB were the first to be informed of interim and final results.

In February 2014, the trial DSMB and Scientific Advisory Board recommended that the results of the first interim analysis should be shared with the international scientific community. These results suggested good tolerance of the drug and the importance of baseline viral load as a key stratification factor in further final analysis, but they did not demonstrate the effectiveness of the drug. However, the research team could not control the interpretation of these interim results by the media and political leaders. Rumors circulated in the national and international media about favipiravir efficacy. With decreasing EVD incidence rates in Forested Guinea, it became obvious that the initial estimation of the patient cohort size would not be met. National authorities strongly urged the trial leaders to publicly announce the results before the trial completion in order to make the drug available in other Guinean ETCs.

Roles and responsibilities

Both MSF and INSERM had little experience in the unique collaboration between a humanitarian organization and a typical life-science research institute.

The trial agreement stipulated that INSERM was responsible for compliance with the trial protocol and the quality of data collection and analysis, while MSF was responsible for patient management. Communication channels were improved when MSF appointed a research coordinator to coordinate trial preparation and patient care and to ensure full integration of the research and clinical teams.

Additional trial staff, financed by INSERM, were recruited with the same working contracts as MSF national staff and trained together with the other ETC staff in the trial SOPs. This

process created a strong team spirit and a feeling of ownership of the project and avoided dividing the staff into 2 groups with different management and working conditions.

Field logistics were provided by MSF, while INSERM was responsible for official communications about the trial with the media and the authorities.

Discussion

Implementing a phase II clinical trial during a devastating epidemic in a research-illiterate environment was full of challenges. Success required cultural sensitivity to build trust within communities and among patients, inclusion criteria based on equity and a risk/benefit assessment, close collaboration between research and clinical teams, and flexibility to adapt trial procedures to field constraints. The main pitfalls were related to protocol compliance: the technical lab problems after the introduction of new tests and the contradiction between study endpoint and clinical discharge criteria. Confidentiality of trial data was not fully controlled and required continuous surveillance.

In a context of fear and mistrust of international actors, community engagement was essential to ensure the appropriate implementation and progression of the trial. A pretrial initiative to inform and involve community leaders; the development of thoughtful, culturally appropriate messages; and a consensual community strategy were essential aspects of this engagement that should not be neglected in any Ebola virus trial.

Scientists have argued for [22–25] and against [26–28] the use of randomized controlled trials (RCTs) to evaluate Ebola therapies. Our paper focused on field implementation, but this was also influenced by the methodological choice of a single-arm design. With patients facing a high probability of mortality and despite the absence of previous efficacy studies, we believe that individual interests must prevail over purity of trial methodology. It would have been unethical and impractical to run an RCT demanding patients to consent to conventional, ineffective care when a potentially beneficial treatment was available [26, 27]. It is noteworthy that virtually all Ebola virus therapy clinical trials have had a single-arm design [28–31]. During the Ebola outbreak, compassionate use of unapproved treatments was criticized because it did not systematically contribute to building knowledge about therapeutic efficacy [23, 27]. In the JIKI trial, the use of favipiravir in pregnant women followed a WHO-recommended protocol for monitored emergency use [32].

In conclusion, the JIKI trial illustrated how an international research team collaborated both with a humanitarian medical organization and local researchers to improve scientific knowledge, create innovative partnerships, promote local capacity building in research, and help EVD patients. Our observations are context specific but may be useful to understand the complexity and challenges of establishing research activities during future international medical emergencies.

Supporting information

S1 Text. Trial protocol. (PDF)S2 Text. CONSORT statement. (PDF)

Acknowledgments

We are very indebted to all the patients who participated in the trial.

We are sincerely grateful to all the staff in Guéckédou ETC for their work and their involvement in the implementation of the JIKI trial. We specially thank Dr. Anthony J. Reid, medical editor at MSF, for his comments on several early versions of this manuscript, which greatly improved this final version. We would also like to show our gratitude to Gregory Anderson, Department of Preventive and Social Medicine of Laval University, for assistance with text editing.

References

- Cenciarelli O, Pietropaoli S, Malizia A, Carestia M, D'Amico F, Sassolini A, et al. Ebola virus disease 2013–2014 outbreak in west Africa: an analysis of the epidemic spread and response. Int J Microbiol. 2015; 2015:769121. https://doi.org/10.1155/2015/769121 PMID: 25852754
- Dallatomasina S, Crestani RA. Ebola outbreak in rural West Africa: epidemiology, clinical features and outcomes. Tropical medicine \& international health: TM \& IH. 2015; 20(4):448–54. https://doi.org/10. 1111/tmi.12454 Dallatomasina2015.
- 3. Sayburn A. WHO gives go ahead for experimental treatments to be used in Ebola outbreak. BMJ. 2014; 349:g5161. https://doi.org/10.1136/bmj.g5161 PMID: 25122051
- Furuta Y, Gowen BB, Takahashi K, Shiraki K, Smee DF, Barnard DL. Favipiravir (T-705), a novel viral RNA polymerase inhibitor. Antiviral Res. 2013; 100(2):446–54. <u>https://doi.org/10.1016/j.antiviral.2013</u>. 09.015 PMID: 24084488
- Gowen BB, Juelich TL, Sefing EJ, Brasel T, Smith JK, Zhang L, et al. Favipiravir (T-705) inhibits Junin virus infection and reduces mortality in a guinea pig model of Argentine hemorrhagic fever. PLoS Negl Trop Dis. 2013; 7(12):e2614. https://doi.org/10.1371/journal.pntd.0002614 PMID: 24386500
- Nagata T, Lefor AK, Hasegawa M, Ishii M. Favipiravir: A New Medication for the Ebola Virus Disease Pandemic. Disaster medicine and public health preparedness. 2014:1–3. <u>https://doi.org/10.1017/dmp.</u> 2014.151 Nagata2014.
- Oestereich L, Ludtke A, Wurr S, Rieger T, Munoz-Fontela C, Gunther S. Successful treatment of advanced Ebola virus infection with T-705 (favipiravir) in a small animal model. Antiviral Res. 2014; 105:17–21. https://doi.org/10.1016/j.antiviral.2014.02.014 PMID: 24583123
- Mendenhall M, Russell A, Smee DF, Hall JO, Skirpstunas R, Furuta Y, et al. Effective oral favipiravir (T-705) therapy initiated after the onset of clinical disease in a model of arenavirus hemorrhagic Fever. PLoS Negl Trop Dis. 2011; 5(10):e1342. https://doi.org/10.1371/journal.pntd.0001342 PMID: 22022624
- Sissoko D, Laouenan C, Folkesson E, M'Lebing AB, Beavogui AH, Baize S, et al. Experimental Treatment with Favipiravir for Ebola Virus Disease (the JIKI Trial): A Historically Controlled, Single-Arm Proof-of-Concept Trial in Guinea. PLoS Med. 2016; 13(3):e1001967. https://doi.org/10.1371/journal. pmed.1001967 PMID: 26930627
- World Health Organization. Handbook for Good Clinical Research Practice (GCP): Guidance for implementation: WHO; 2002. http://apps.who.int/prequal/info_general/documents/gcp/gcp1.pdf.
- Fairhead J. Understanding social resistance to Ebola response in Guinea. Ebola Response Anthropology Platform, 2015.
- 12. Anoko J. Communication with rebellious communities during an outbreak of Ebola Virus Disease in Guinea: an anthropological approach. Ebola Response Anthropology Platform, 2015.
- Carrion Martin AI, Derrough T, Honomou P, Kolie N, Diallo B, Kone M, et al. Social and cultural factors behind community resistance during an Ebola outbreak in a village of the Guinean Forest region, February 2015: a field experience. Int Health. 2016; 8(3):227–9. https://doi.org/10.1093/inthealth/ihw018 PMID: 27059272
- 14. World Medical Association. Declaration of Helsinki. 2008.
- 15. National Commission for the Protection of Human Subjects of Biomedical and Behavioral Research. The Belmont Report. U.S. Department of Health, Education, and Welfare, 1976.
- Agua-Agum J, Ariyarajah A, Blake IM, Cori A, Donnelly CA, Dorigatti I, et al. Ebola virus disease among children in West Africa. The New England journal of medicine. 2015; 372(13):1274–7. https://doi.org/ 10.1056/NEJMc1415318 Agua-Agum2015.
- Mupere E, Kaducu OF, Yoti Z. Ebola haemorrhagic fever among hospitalised children and adolescents in northern Uganda: epidemiologic and clinical observations. African health sciences. 2001; 1(2):60–5. Mupere2001. PMID: 12789118
- Baron RC, McCormick JB, Zubeir OA. Ebola virus disease in southern Sudan: hospital dissemination and intrafamilial spread. Bull World Health Organ. 1983; 61(6):997–1003. PMID: 6370486

- Bouazza N, Treluyer JM, Foissac F, Mentre F, Taburet AM, Guedj J, et al. Favipiravir for children with Ebola. Lancet. 2015; 385(9968):603–4. <u>https://doi.org/10.1016/S0140-6736(15)60232-X</u> PMID: 25706078
- Black BO, Caluwaerts S, Achar J. Ebola viral disease and pregnancy. Obstet Med. 2015; 8(3):108–13. https://doi.org/10.1177/1753495X15597354 PMID: 26457118
- Bebell LM, Riley LE. Ebola virus disease and Marburg disease in pregnancy: a review and management considerations for filovirus infection. Obstet Gynecol. 2015; 125(6):1293–8. https://doi.org/10.1097/ AOG.00000000000853 PMID: 26000499
- 22. Cox E, Borio L, Temple R. Evaluating Ebola therapies—the case for RCTs. N Engl J Med. 2014; 371 (25):2350–1. Epub 2014/12/04. https://doi.org/10.1056/NEJMp1414145 PMID: 25470568
- 23. Joffe S. Evaluating novel therapies during the Ebola epidemic. Jama. 2014; 312(13):1299–300. Joffe2014. https://doi.org/10.1001/jama.2014.12867 PMID: 25211645
- 24. Lane HC, Marston HD, Fauci AS. Conducting clinical trials in outbreak settings: Points to consider. Clin Trials. 2016; 13(1):92–5. https://doi.org/10.1177/1740774515618198 PMID: 26768564
- Lanini S, Zumla A, Ioannidis JP, Di Caro A, Krishna S, Gostin L, et al. Are adaptive randomised trials or non-randomised studies the best way to address the Ebola outbreak in west Africa? Lancet Infect Dis. 2015; 15(6):738–45. <u>https://doi.org/10.1016/S1473-3099(15)70106-4</u> PMID: <u>25881871</u>
- Adebamowo C, Bah-Sow O, Binka F, Bruzzone R, Caplan A, Delfraissy J-Fc, et al. Randomised controlled trials for Ebola: practical and ethical issues. The Lancet. 2014; 384(9952):1423–4. <u>https://doi.org/10.1016/S0140-6736(14)61734-7</u> Adebamowo2014.
- Calain P. The Ebola clinical trials: a precedent for research ethics in disasters. J Med Ethics. 2016. https://doi.org/10.1136/medethics-2016-103474 PMID: 27573153
- Whitehead J, Olliaro P, Lang T, Horby P. Trial design for evaluating novel treatments during an outbreak of an infectious disease. Clin Trials. 2016; 13(1):31–8. <u>https://doi.org/10.1177/1740774515617740</u> PMID: 26768556
- 29. van Griensven J, Edwards T, de Lamballerie X, Semple MG, Gallian P, Baize S, et al. Evaluation of Convalescent Plasma for Ebola Virus Disease in Guinea. N Engl J Med. 2016; 374(1):33–42. <u>https://doi.org/10.1056/NEJMoa1511812 PMID: 26735992</u>
- Dunning J, Kennedy SB, Antierens A, Whitehead J, Ciglenecki I, Carson G, et al. Experimental Treatment of Ebola Virus Disease with Brincidofovir. PLoS ONE. 2016; 11(9):e0162199. <u>https://doi.org/10.1371/journal.pone.0162199 PMID: 27611077</u>
- Dunning J, Sahr F, Rojek A, Gannon F, Carson G, Idriss B, et al. Experimental Treatment of Ebola Virus Disease with TKM-130803: A Single-Arm Phase 2 Clinical Trial. PLoS Med. 2016; 13(4): e1001997. https://doi.org/10.1371/journal.pmed.1001997 PMID: 27093560
- World Health Organization. Ethical issues related to study design for trials on therapeutics for Ebola Virus Disease: WHO Ethics Working Group meeting 20–21 October, summary of discussion. Geneva: WHO, 2014.